Author’s response to reviews

Title: Case report: Prolonged and high dosage of tigecycline: Successful treatment of spondylodiscitis caused by multidrug resistant Acinetobacter baumannii

Authors:

Olga Tsachouridou (OlgaT_med@hotmail.com)
Adamantini Georgiou (adamantin@auth.gr)
Sideris Nanoudis (sidnanoudis@yahoo.gr)
Theofilos Chrysanthidis (teochriss@yahoo.co.uk)
Georgia Loli (loligeorgia.med@gmail.com)
Petros Morfesis (petrosmed1@gmail.com)
Pantelis Zebekakis (pzempeka@auth.gr)
Symeon Metallidis (metallidissimeon@yahoo.gr)

Version: 2 Date: 18 May 2017

Author’s response to reviews:

Page 2, line 2
Grammatical error corrected (highlighted)

Page 2, line 9
The patient denied surgical treatment (highlighted)

Page 3, line 15
Spelling error corrected (highlighted)
Page 4, line 3
Grammatical error corrected (highlighted)

Page 4, line 13
Levofloxacin 500mg orally twice daily for 7 more days (highlighted)

Page 5, line 2
Colistin (highlighted)

Page 5, line 13
Grammatical error corrected (highlighted)